Author's response to reviews

Title: Bath Breakfast Project (BBP) - Examining the Role of Extended Daily Fasting in Human Energy Balance and Associated Health Outcomes: study protocol for a randomised controlled trial [ISRCTN31521726].

Authors:

James A Betts (j.betts@bath.ac.uk)
Dylan Thompson (d.thompson@bath.ac.uk)
Judith D Richardson (j.d.richardson@bath.ac.uk)
Enhad A Chowdhury (e.chowdhury@bath.ac.uk)
Matthew Jeans (m.jeans@bath.ac.uk)
Geoffrey D Holman (g.d.holman@bath.ac.uk)
Kostas Tsintzas (kostas.tsintzas@nottingham.ac.uk)

Version: 4 Date: 8 July 2011

Author's response to reviews:

We have picked-up very minor typographical or formatting errors on pages 9, 10 and 13 of the main document. These are identified in red text on the attached version.

We also noticed a minor error on one of the two figures. Specifically, figure 2 indicated that we would aim for a 1:1 distribution of habitual breakfast consumers/skippers in the phase II allocation, whereas this is not the case.